

GOPEN ACCESS

Citation: Liu J, Yue M, Yu F, Tong Y (2022) The contribution of tourism mobility to tourism economic growth in China. PLoS ONE 17(10): e0275605. https://doi.org/10.1371/journal.pone.0275605

Editor: Hironori Kato, The University of Tokyo, JAPAN

Received: March 3, 2022

Accepted: September 20, 2022

Published: October 27, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0275605

Copyright: © 2022 Liu et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data on RAILWAY, HIGHWAY, ROAD1, ROAD2, ROAD, GDP, TERTIARY INDUSTRY, and POPULATION are from the Chinese Nation Bureau of Statistics (https://data.stats.gov.cn/easyquery.htm?cn=C01). **RESEARCH ARTICLE**

The contribution of tourism mobility to tourism economic growth in China

Jun Liu¹, Mengting Yue², Fan Yu^{3*}, Yun Tong⁴

 School of Tourism, Hubei University, Wuhan, Hubei, China, 2 School of Urban and Regional Science, East China Normal University, Shanghai, China, 3 School of Business, Hubei University, Wuhan, Hubei, China,
 School of Tourism, Hainan University, Haikou, Hainan, China

* 201901111200032@stu.hubu.edu.cn

Abstract

Mobility is the key factor in promoting tourism economic growth (TEG), and the transportation infrastructure has essential functions for maintaining an orderly flow of tourists. Based on the theory of fluid mechanics, we put forward the indicator of tourism mobility (TM). This study is the first to measure the level of TM in China and analyze the spatiotemporal evolution characteristics of TM. Applying the Exploratory Spatial Data Analysis method, we analyze the global and local spatial correlation characteristics of TM. Moreover, we further estimate the contribution of TM to TEG by econometric models and the LMDI method. The results show that (1) the TM in China has maintained rapid growth for a long time. However, there are differences in the rate of growth in different regions. The TM in each region only showed a significant positive spatial correlation in 2016–2018. The space-time pattern is constantly changing over time. The local spatial autocorrelation results of TM are stable, and various agglomeration states are stably distributed in some provinces. (2) The regression results of the traditional panel data model and spatial panel data model both show that TM has a significant positive effect on TEG. Moreover, TM has a negative spatial spillover effect on neighboring regions. (3) The result from the decomposition of LMDI shows that the overall contribution of TM to TEG is 15.76%. This shows that improving TM is a crucial way to promote the economic growth of tourism.

Introduction

In recent years, the tourism industry has maintained rapid development. By 2019, the total number of global tourist trips exceeded 12.3 billion, an increase of 4.6% over the previous year. The total global tourism revenue was US\$5.8 trillion, equivalent to 6.7% of global GDP (World Tourism Economy Trends Report [1]). Tourism has made important contributions to economic growth by increasing employment, improving infrastructure, and accumulating foreign exchange earnings for destinations [2]. Due to the impact of COVID-19, People's travel is restricted. The total number of international tourists in 2021 decreased by 72% compared with 2019, and international tourism consumption dropped by nearly half compared with 2019 [3].

The above facts remind us that mobility has become an essential feature of tourism activities [4, 5]. Tourists from origins to destinations result in a series of mobility of information,

The data on TOURISM REVENUE and VISITORS are from the CEIC database (https://insights. ceicdata.com). The data on TRAFFIC, TOURISM MOBILITY, RECPTION, INDUSTRY, and STRUCTURE were calculated by the authors. Please see the paper for details.

Funding: This work was supported by grants from National Social Science Foundation of China [grant number 17CJY051].

Competing interests: The authors have declared that no competing interests exist.

material, and capital. These mobilities have a great influence on tourist destinations [6–9]. If tourism mobility (TM) stagnates, tourist attractions, reception facilities and transportation facilities built for tourists will be idle. Tourism workers will lose their jobs and tourism economic growth (TEG) will also stagnate. Therefore, studying the impact of TM is necessary and important.

As one of the important tourist destinations in the world, China's domestic tourism and inbound tourism are developing rapidly. In 2019, the total contribution of China's tourism industry to GDP reached 10.94 trillion yuan, accounting for 11.05% of the total GDP, exceeding the proportion of international tourism in the global GDP. A total of 28.25 million people were directly employed in tourism, and 51.62 million people were indirectly employed in tourism. The total employment in tourism accounts for 10.31% of the total employed population in the country [10]. However, due to the impact of COVID-19, the development level of China's tourism industry has not recovered to the level of 2019. In 2021, the total number of domestic tourists in China was 3.246 billion, which is only 54% of that in 2019, and directly leads to a total tourism revenue of 2.92 trillion yuan, which is only 51% of that in 2019. This shows that TM is more important to China's tourism industry. Therefore, we decide to focus on the TM in this study and take China as the research sample.

The top priority of this study is to obtain the right measurement of TM. Transportation infrastructure is an important carrier for the exchange of factors in tourism. Existing studies have confirmed that transportation is a key factor in promoting TEG [11–13]. The establishment of the transportation system has an obvious effect on improving the accessibility of tourist destinations and promoting the inflow of the tourist population [14]. However, most existing studies only take tourist arrivals to characterize TM [15–21]. They ignore that the transportation infrastructure is also an important factor affecting the TEG. Therefore, this study redefines TM, which considers both transport infrastructure and tourist arrivals.

Another important purpose of this study is to explore the effect of TM on TEG. Existing literature analyzes the links between TM and international trade [22, 23] or focuses on the relationship between economic growth [24, 25]. However, less literature has focused on the relationship between TM and TEG. There are two possible reasons for the lack of attention. First, the positive and significant impact of the tourist arrivals and TEG no longer needs to be verified. It is common sense that the more tourists the destination receive, the higher the tourism income. Second, tourist arrivals, as a single indicator to measure TM, are able to affect the TEG. Our measurement of the TM concludes both transport infrastructure and tourist arrivals in this study. Therefore, we decide to explore the contribution of TM to the TEG based on the new measurement for TM.

We first use econometric methods to test whether there is a significant impact of TM on TEG. Considering the positive impact of transport infrastructure on China's TEG [26], we hypothesize that TM has a positive impact on TEG. Previous studies have also shown that the spatial spillover effect of tourism may significantly affect the TEG [27–29]. Therefore, we further apply the spatial Durbin model to test the impact of TM on TEG.

Moreover, we also use the LMDI (Logarithmic Mean Divisia Index) method to further analyze the contribution of TM to TEG in more detail. The LMDI method is often used to study environmental issues such as energy consumption and carbon emissions [30, 31]. In the field of tourism research, the LMDI method is mostly used to decompose tourism carbon emissions or energy consumption [32, 33]. Few studies are using the LMDI to analyze TEG. Therefore, we further use the LMDI method to decompose TEG into five influencing factors including the tourism mobility effects (*TM*), the cumulative traffic effects (*Traffic*), the effects of the tertiary industry (*Industry*), the structural effects of the tourism industry (*Structure*) and the reception effects (*Reception*), and examine the contribution of TM to TEG. Different from previous studies, this study makes two contributions to the literature. First, we introduce the related concepts of fluid mechanics to construct the indicator TM. We also consider the superposition effect of tourist arrivals and transportation infrastructure. This deepens the understanding of TM and promotes the integration of interdisciplinary knowledge. Second, we are the first to examine the impact of TM on TEG using econometric models and the LMDI method. This deepens the understanding of the mechanisms that influence TEG. The results of this study also provide a reference for tourism-related policy makers. Regions wishing to develop tourism can achieve TEG by expanding the size of the source market and promoting the construction of transportation infrastructure.

The rest of this study is organized as follows. Section 1 summarizes the relevant literature. Section 2 presents the theoretical framework, methods, and data. Section 3 introduces the spatiotemporal pattern and evolutionary trend of TM. Section 4 analyzes the contribution of TM to TEG from two different perspectives. Section 5 discusses and analyzes the research results. The last section concludes this study.

Literature review

As the core of tourism activities, TM refers to the mobility of tourists from the origin to the destination, and the stay of tourists in the region [34]. It is often associated with tourism demand and is measured by tourist arrivals [35]. Since the 1970s, many studies have paid attention to the influencing factors and the spatial structure of TM [15, 16]. The existence of regional heterogeneity makes TM affected by many factors, such as infrastructure, income, GDP, and cultural distance [17, 18, 20]. Moreover, it also makes the spatial structure of TM different. Therefore, TM prediction has become one of the research hotspots [36]. A large body of research has focused on TM forecasting [21], including using a combination and integration of forecasts, using nonlinear methods for forecasting, and extending existing methods to better model the changing nature of tourism data [37]. The gravity model is an earlier method used to analyze international TM [38]. Due to its effectiveness in explaining TM [22], gravity models are often used to analyze international tourism service trade. Although the use of gravity models to predict bilateral TM still lacks a corresponding theoretical explanation mechanism, empirical evidence supports the applicability and robustness of gravity models for TM [23]. Existing research focuses on examining the movement patterns and spatial structure of international TM in destinations [39], such as the transfer of inbound TM within regions and the influencing factors of inbound TM within destinations [40]. There are still few studies on the overall spatial characteristics of TM within destination countries, and the only literature is mainly based on digital footprints or questionnaire data to analyze the spatial structure of TM [41, 42].

Unlike the tourist arrivals indicator, which focuses more on the mobility of people, TM examines a wider range of content, including the mobility of people, the mobility of materials, the mobility of ideas (more intangible thoughts and fantasies), and the mobility of technology [8]. The early tourist movement focused more on tourist travel decisions and the resulting movement patterns. Lue et al. [43] summarized five travel patterns of tourists between destinations. Li et al. [44] revealed the spatial patterns of TM and tourism propensity in the Asia-Pacific region over the past 10 years. McKercher and Lau [45] took Hong Kong as an example and identified 78 movement patterns and 11 movement styles of TM within the destination. In recent years, with the help of technologies such as GPS, GIS, and RFID, the movement in national parks, theme parks, protected areas, etc. continues to increase [47–49], and explore the influencing factors of visitor movement [50], broadening the microscale visitor mobility research

content. TM also has economic, social, and cultural impacts on destinations through the movement of tourists. Numerous empirical studies have shown that tourist arrivals have a positive impact on economic growth [51]. Tourism is an important driver of economic growth [52]. However, some studies have shown that tourist arrivals do not directly lead to economic growth, but promote TEG through regional economic development [53–55]. The mobility of tourism will also bring about changes in destination transportation facilities. Transportation is not only an important carrier of TM but also an important part of tourists' travel experience [8]. It also has a positive impact on destination company value together with TM [26].

There are many theoretical discussions and empirical studies on the factors influencing TEG. From the perspective of suppliers, resource endowment [56–58] and environmental quality [59–62] are the fundamental factors determining tourism development. Simultaneously, as a typical service industry, human capital and physical capital in the tourism industry [63, 64] and service level [65] will impact tourism economic efficiency. From the perspective of demanders, the rise of per capita income and consumption upgrading continue to drive the transformation in the tourism industry [66], which in turn leads to an increasing scale of market demand [67], which provides the possibility of increasing the foreign exchange earnings, local capital accumulation, and consumption spillovers. From the perspective of supporters, scholars have verified the significant effects of factors on TEG, including the transportation facilities and accessibility [68–71], the basis of the economy and marketization [72], industrial structure [73], public policy [74–76], and technological progress [77].

In summary, the research on TM has paid attention to its impact on the regional economy, but they both ignored the role of TM on TEG. Studies of TEG based on static factors have primarily relied on econometric models [78]. Although the spatial spillover effects of influencing factors have gradually gained attention, its depth is limited and fails to explore the impact of TM and other related factors on the TEG. TM is becoming central to tourism activities and understanding the capital mobility of tourism will have implications for tourism development under the new mobility paradigm [79]. This study proposes the concept of TM based on the theory of fluid mechanics, explores its impact on TEG, and analyzes the contribution of each influencing factor to TEG.

Theoretical framework, research methods, and data sources Theoretical framework

Traditionally, tourism research considers the tourism system as tourist sources, tourist destinations, and tourist corridors (transportation systems) [80, 81]. Under the new mobility paradigm, this study regards the spatial transfer of tourists from the source to the destination as a mobility process. Tourist mobility is the fundamental reason for the existence of tourism. If tourists stop flowing, tourism will cease to exist.

It is known that the fluid will be affected by a variety of factors, such as viscosity, density, resistance coefficient, and altitude. As shown in Fig 1, the total mobility of tourists from a tourist origin to a tourist destination is the number of tourists (Q). The spatial transfer of tourists, on the other hand, requires the use of transportation infrastructure as well as means of delivery. As an essential vehicle to support tourism development, transportation infrastructure directly reflects regional accessibility and relevance and is a crucial factor influencing TM [82–84], and its construction level has different effects on TEG in different regions [11, 85–87]. According to the equations in fluid mechanics, the average velocity is equal to the flow rate ratio to the cross-sectional area. It can be deduced that TM = Q/TL. TM is determined by the number of tourists (Q) and the length of transportation infrastructure (TL). According to the definition, this indicator considers both tourist arrivals and flow rate, and its significance lies



https://doi.org/10.1371/journal.pone.0275605.g001

in its ability to characterize the mobility of tourism factors relying on tourists and physical transportation. This paper also connects the factor decomposition method to determine the importance of TM to TEG and presents theoretical implications for identifying essential factors to enhance tourism efficiency and stimulate tourism industry development.

Research methods

Measurement of tourism mobility. The basic principle of fluid mechanics is that the average velocity is proportional to the flow rate and inversely proportional to the cross-sectional area. This paper characterizes the flow rate by the number of tourist inflows, and the length of transportation infrastructure represents the cross-sectional area, with the equation as (1)

$$\Gamma M_{i}^{t} = Q_{i}^{t} / T L_{i}^{t}$$

$$\tag{1}$$

where TM_i^t represents the TM in province i and year t; Q_i^t is the number of tourists in province i and year t; and TL_i^t is the length of the weighted transportation infrastructure in province i and year t, including railroads, highways, primary roads, secondary roads, and other grades of roads. China's railway and road passenger traffic accounts for the vast majority of the total passenger traffic. Furthermore, we were unable to calculate weighted air and water transportation infrastructure lengths, so we only consider the land transportation infrastructure data. The length of transportation infrastructure is weighted according to Chen et al. [88].

Exploratory spatial data analysis. It is generally believed that tourism has a spatial spillover effect and spatial correlation [28]. Therefore, we use Exploratory Spatial Data Analysis (ESDA) to detect spatial correlation among the variables. ESDA is used to analyze spatial characteristics through global and local spatial autocorrelation measurements [42, 89].

The global Moran's I is an indicator of whether factors are spatially correlated and its value ranges from -1 to 1. When $0 < I \le 1$, it indicates a positive spatial correlation; when $-1 \le I < 0$, it indicates a negative spatial correlation; when I = 0, there is no spatial relationship. The equation is as in (2).

$$I = \frac{n \sum_{i=1}^{n} \sum_{j=1}^{n} W_{ij} |TEG_i - \overline{TEG}| |TR_j - \overline{TEG}|}{\sum_{i=1}^{n} \sum_{j=1}^{n} W_{ij} \sum_{i=1}^{n} |TR_i - \overline{TEG}|^2}$$
(2)

Where TEG_i and TEG_j denote the tourism revenue of provinces i and j, respectively; n is the number of provinces; $\overline{\text{TEG}}$ denotes the average value of tourism revenue of each province; W_{ij} represents the spatial weight matrix of provinces i and j. We choose the adjacency matrix and use Guangdong and Guangxi as the neighboring provinces of Hainan.

Local spatial autocorrelation is used to explore cluster patterns and spatial patterns [90, 91]. We analyze the local spatial autocorrelation characteristics through cluster and outlier analyses. The calculation process is expressed as Formula (3):

$$I_p = \frac{(n-1)\sum_{q=1,q\neq p}^{n} W_{pq}(TEG_p - \overline{TEG})(TEG_q - \overline{TEG})}{\sum_{q=1,q\neq p}^{n} (TEG_q - \overline{TEG})^2}$$
(3)

With a Z statistical test as in Formula (4), the cluster and outlier analyses can identify H_H (High_High) clusters, L_L (Low_Low) clusters, L_H (low value surrounded by high values) clusters, and H_L (high value surrounded by low values) clusters at a 95% confidence level.

$$Z_{I_p} = \frac{I_p - E[I_p]}{\sqrt{V[I_p]}} \tag{4}$$

Econometric model. The econometric model, including tourism economic growth (TEG), tourism mobility (TM), physical capital in the tourism industry (TP), and human capital in the tourism industry (TH), is constructed according to economic growth theory without considering spatial spillover effects. Besides, since the measurement of TM only considers land transportation infrastructure data, the passenger traffic by the airport (TA) is introduced in the model to characterize the air capacity. Eq (5) represents the econometric model (TEG_{it}) in province i and year t, where α is the constant term, β is the parameter to be estimated, μ_i denotes the spatial effect, and ε_{it} denotes the random error term.

$$Ln(TR)_{it} = \alpha + \beta Ln(TM)_{it} + \beta_1 Ln(TP)_{it} + \beta_2 Ln(TH)_{it} + \beta_3 Ln(TA)_{it} + \mu_i + \varepsilon_{it}$$
(5)

However, the spatial correlation of TEG will lead to biased parameter estimates of traditional econometric models. If the test results of global Moran's I indicate that TEG is significantly spatially correlated, a spatial econometric model should be introduced to solve the biasvariance problem. The spatial Durbin model (Eq.6) is developed according to Eq.5. The spatial weight matrix used in the spatial Durbin model is an adjacency matrix. y_{it} represents the TEG in province i and year t; x_{it} represents the TM, TP, TH, and TA in province i and year t; and $W_{ij}y_{jt}$ and $W_{ij}x_{jt}$ are the TEG and lagged terms of each influencing factor, respectively. ρ and ϕ are spatial lagging coefficients, and v_t denotes the time effect.

$$\mathbf{y}_{it} = \rho \sum_{j=1}^{n} \mathbf{W}_{ij} \mathbf{y}_{jt} + \beta \mathbf{x}_{it} + \sum_{j=1}^{n} \varphi \mathbf{W}_{ij} \mathbf{x}_{jt} + \mu_{i} + \mathbf{v}_{t} + \boldsymbol{\varepsilon}_{it}$$
(6)

LMDI decomposition. The LMDI decomposition method is widely used because it can effectively solve the residual problem in the decomposition and zero and negative values in the data. LMDI In this study, TEG is decomposed according to Eq (7). The influencing factors of TEG are decomposed into tourism mobility effects (*TE*), cumulative traffic effects (*Traffic*), effects of the tertiary industry (*Industry*), structural effects of the tourism industry (*Structure*), and reception effects (*Reception*). The equations are shown in (8) to (11). Traffic indicates the weighted road length; GDP (service) intimates the value added of the tertiary industry; Population represents the population in each province, and Visitors is the number of tourists. Introducing the log-average function L(x,y) defined in Eq (12). Eq (7) is decomposed into Eq (13) by LMDI, where Δ TEG denotes the amount of change in TEG from initial time 0 to period t, and Δ TM, Δ T, Δ I, Δ S, Δ W represent the contribution of each influencing factor to TEG. The

equations are shown in (14) to (18).

$$TEG = TE * Traffic * Industry * Structure * Reception$$
(7)

$$TM = Visitors/Traffic$$
 (8)

$$Industry = GDP(service) / Population$$
(9)

$$Structure = TEG/GDP(service)$$
(10)

$$Reception = Population/Visitors$$
(11)

$$L(x, y) = \begin{cases} \frac{x - y}{\ln x - \ln y}, & x \neq y \\ x, x = y \\ 0, & x = y = 0 \end{cases}$$
(12)

$$\Delta TEG = \Delta TEG(t) - \Delta TEG(0) = \Delta TM + \Delta T + \Delta I + \Delta S + \Delta W$$
(13)

$$\Delta TM = \frac{\Delta TEG(t) - \Delta TEG(0)}{\ln TEG(t) - \ln TEG(0)} * \ln \left(\frac{TM(t)}{TM(0)}\right)$$
(14)

$$\Delta T = \frac{\Delta TEG(t) - \Delta TEG(0)}{\ln TEG(t) - \ln TEG(0)} * \ln \left(\frac{Traffic(t)}{Traffic(0)}\right)$$
(15)

$$\Delta I = \frac{\Delta TEG(t) - \Delta TEG(0)}{\ln TEG(t) - \ln TEG(0)} * \ln \left(\frac{\text{Industry}(t)}{\text{Industry}(0)}\right)$$
(16)

$$\Delta S = \frac{\Delta TEG(t) - \Delta TEG(0)}{\ln TEG(t) - \ln TEG(0)} * \ln \left(\frac{Structure(t)}{Structure(0)}\right)$$
(17)

$$\Delta W = \frac{\Delta TEG(t) - \Delta TEG(0)}{\ln TEG(t) - \ln TEG(0)} * \ln \left(\frac{\text{Reception}(t)}{\text{Reception}(0)}\right)$$
(18)

Data sources

The study area is 31 provinces of China (excluding Hong Kong, Macao, and Taiwan), which is divided into seven regions according to the geographical divisions of China. The provinces included in each region are listed in supporting information. Since data availability varies widely across regions, the research period of TM and LMDI decomposition is from 2000 to 2018. As the National Bureau of Statistics of China (NBS) started to collect the employment data of private enterprises and individuals by sector in 2004 and the data for 2018 has not been updated yet, the research period of the spatial econometric model only covers the period from 2004 to 2017.

The data sources involved in the paper are as follows: the transportation infrastructure data come from the China Statistical Yearbook; the number of tourists is obtained from the Statistical Bulletin on National Economic and Social Development. Air passenger traffic data is collected from Civil Aviation Airport Production Statistics Bulletin. We employ the social fixed asset investment in transportation, storage, and postal services, wholesale and retail trade,

accommodation and catering, and culture, sports, and entertainment as proxies for physical capital in the tourism industry (TP). This is because various aspects influence tourism development. Considering that only direct tourism investment does not reflect the total investment in tourism by society, we choose the four industries closely related to tourism development as physical capital in the tourism industry.

In this paper, private and individual employees in the transport, storage, and postal industry, wholesale and retail trade, and accommodation and catering industries are used to represent the human capital in the tourism industry (TH). The main reason for this is that, on the one hand, most studies only consider the number of employees in travel agencies, scenic spots, and star hotels, which differs significantly from the actual number of direct and indirect employees in tourism. On the other hand, since private enterprises and individual employment solve more than 80% of the urban employment problem, the number of private enterprises and individual employment in the three industries related to the tourism industry is chosen to represent the human capital. All the above data are collected from the NBS (http:// data.stats.gov.cn). In the LMDI decomposition, the value added of the tertiary industry and the population in each province come from the China Statistical Yearbook.

Analysis of tourism mobility measurement results

Spatiotemporal evolution characteristics of tourism mobility

Limited by space, Table 1 only shows the results of TM over five years. During the study period, TM increased from 56~12745 p visitors /km to 382~18865 p visitors /km, with an average annual growth rate between 2.20% and 13.46%. According to the average value of TM (Fig 2), the study areas are divided into the following three types.

- 1. "Leading Area", including East China and North China, ranked first and second in all regions. Their TM increased from 2679.39 and 1884.34 p visitors/km in 2000 to 5859.93 and 5209.94 p visitors/km in 2018. However, their annual average growth rates were 5.07% and 6.43%, respectively, ranking first and second from the bottom in all regions. East China is located on the coast, relying on superior natural conditions and an economic foundation, and its regional transportation system is relatively complete. Therefore, it has formed many advantageous tourist resource gathering areas and has become the main tourist destination of inbound tourists in China, and its mobility has long ranked first in the country. As a political and economic center, Beijing has become a tourist attraction for domestic and inbound tourism with a large number of historical and cultural tourism resources. It also drives the joint development of the tourism industry in North China with the Beijing-Tianjin-Hebei urban agglomeration as the core, making North China the second largest core area of TM after East China.
- 2. "Stable Area", including South China, Southwest China, Central China, and Northeast China, ranked third to sixth in all regions. Their TM increased from 903.57p visitors/km, 695.15p visitors/km, 632.06p visitors/km, 493.33 p visitors/km in 2000 to 2626.11p visitors/km, 2754.97p visitors/km, 2857.88p visitors/km, 2244.68 p visitors/km in 2018. The average annual growth rates were 6.58%, 8.81%, 9.06%, and 9.38%, respectively. TM in South China grew rapidly during 2005~2015, while it has gradually slowed down in recent years. This is mainly due to the construction of the early transportation system in South China, which increased tourist mobility. After the basic construction of facilities, the incremental tourist inflows decreased, and the overall growth remained stable. Central China has become one of the core transportation hubs under its location and has driven regional tourism development, becoming a central province in the second echelon of TM. Due to geographical

Table 1. Evaluation of TM (unit: p visitor/km).

Provinces	2000	2005	2010	2015	2018
Beijing	5815.61	6244.13	6396.40	8802.59	9749.22
Tianjin	2423.53	3244.66	2806.19	7044.53	9087.30
Hebei	643.89	861.74	808.73	1635.52	2808.76
Shanxi	442.66	751.96	848.94	2084.30	3911.71
Inner Mongolia	95.99	221.08	248.93	380.48	492.70
Liaoning	686.93	1284.09	2478.02	2269.75	4008.91
Jilin	385.53	441.30	627.62	1202.56	1710.53
Heilongjiang	407.54	532.26	1093.06	760.01	1014.61
Shanghai	12745.08	8117.20	13246.92	15309.97	18864.86
Jiangsu	1992.54	1819.48	2047.88	3226.98	4053.52
Zhejiang	1347.43	2177.08	2358.99	3762.31	4609.22
Anhui	573.09	576.14	951.49	2089.90	3015.35
Fujian	639.86	1001.07	1339.40	2378.62	3825.36
Jiangxi	703.39	866.53	879.65	2419.10	4047.48
Shandong	754.32	1269.68	1317.57	2116.74	2603.69
Henan	701.00	974.92	1182.94	2121.21	2715.03
Hubei	922.82	807.60	977.49	1811.08	2295.51
Hunan	1086.90	1183.50	1001.53	1922.76	2876.79
Guangdong	694.74	863.53	1941.69	3419.35	1874.52
Guangxi	744.77	1013.08	1404.58	2434.53	4532.18
Hainan	645.95	987.93	1007.22	1648.52	1858.19
Chongqing	1214.23	1552.96	1798.32	3056.53	3919.89
Sichuan	654.32	1346.55	1176.15	1966.18	2027.73
Guizhou	878.41	710.70	1540.86	2521.01	5206.88
Yunnan	357.47	567.94	804.54	1465.18	2753.10
Tibet	55.86	159.35	178.99	326.26	381.81
Shaanxi	671.95	944.96	918.24	2108.71	3256.66
Gansu	198.24	279.20	428.90	1089.77	1925.55
Qinghai	165.60	194.06	216.32	284.30	474.31
Ningxia	191.20	277.46	349.91	416.67	704.38
Xinjiang	183.04	175.42	253.72	354.96	777.00



Fig 2. Mean value of regional TM from 2000 to 2018.

https://doi.org/10.1371/journal.pone.0275605.g002

restrictions, Northeast and Southwest China are less connected to the transportation network than coastal areas, resulting in relatively low levels of TM. Northeast China focuses on the development of heavy industry but pays little attention to the tertiary industry, and tourism infrastructure construction and resource development are relatively weak, which leads to low TM. There are many mountains in Southwest China, and its early traffic development level lags. With the opening of the Chengdu-Chongqing high-speed railway and Chengdu-Guizhou high-speed railway, and the development of the air transportation industry, the land and air transportation layout in Southwest China is becoming increasingly mature. Southwest China actively developed its resources, and the tourist inflow increased from 145 million (2000) to 2.994 billion (2018), with an average value of TM catching up with that of southern China during 2016~2018.

3. "Potential Area", including Northwest China, ranks last in terms of average tourist mobility. Its TM increased from 282.01 p visitors/km in 2000 to 1427.58 p visitors/km in 2018, but its average annual growth rate was 10.01%, ranking first among all regions. As less developed region, Northwest China has a poor foundation in economic development and openness to the outside world, and TM has long been at the bottom of the list. Although TM in Northwest China has long been at the bottom of the list, its mobility growth rate leads other regions as tourism infrastructure construction and resource development levels have improved under the active promotion of Western Development policies, the Five-Year Plan, and the Territorial Tourism Strategy.

To more intuitively observe the temporal and spatial change characteristics of TM during the study period, we apply the method of natural breaks to classify the 31 provinces. Natural breaks classes are based on natural groupings inherent in the data. Class breaks are identified that best group similar values and maximize the differences between classes. The features are divided into classes whose boundaries are set where there are relatively big differences in the data values. The natural breaks classification method is a data classification method designed to determine the best arrangement of values into different classes. This is done by seeking to minimize each class's average deviation from the class mean while maximizing each class's deviation from the means of the other groups [92]. We divided the 31 provinces into five categories, highest-value area, higher-value area, medium-value area, lower-value area, and lowestvalue area, according to the TM in 2000, 2005, 2010, 2015, and 2018. As shown in Fig 3, (1) Shanghai and Beijing have long been in the highest-value area and higher-value area of TM. Tibet, Qinghai, Ningxia, Xinjiang, Inner Mongolia, Gansu, Jilin, Heilongjiang, Hubei, and Hainan have long been in the lowest-value and lower-value areas. (2) Over time, the number of provinces in the highest-value area and the higher-value area increased significantly, from 2 provinces in 2000 to 12 provinces in 2018. The number of provinces in the lowest-value area and lower-value area significantly decreased, from 26 provinces in 2000 to 12 provinces in 2018; the number of provinces in the medium-value area fluctuated randomly, with the fewest 3 in 2000 and the most 13 in 2015. (3) Except for Shanxi, Northwest China has been in the lowest-value area and the lower-value area for a long time; The TM values in Southwest China have changed greatly. Chongqing and Guizhou have jumped from the lower-value area to the higher-value area, and Yunnan has jumped from the low-value area to the medium-value area. Tibet is relatively stable and has been in the lowest-value area for a long time; South China is relatively stable, but the average value TM in Guangxi has changed greatly, jumping from the lower-value area to the higher-value area; The average TM in Central China has been in the low-value area for a long time. Central China is also relatively stable, and its average TM has long been located in the lower-value area and the medium-value area. Except for Shanghai, which has always been in the highest-value area, the initial value of TM in other provinces in



Fig 3. The spatiotemporal pattern and direction distribution of provincial TM. a. 2000, b. 2005, c. 2010, d. 2015, e. 2018.

https://doi.org/10.1371/journal.pone.0275605.g003

East China has jumped upward. In the Northeast, Liaoning's TM has always been in a leading position, and it has gradually transitioned from a lower-value area to a higher-value area. However, Jilin and Heilongjiang have always been in the lowest-value area and the lower-value area, respectively. Changes in TM in North China are diverse. Beijing has long been located in the highest-value area and higher value area. Inner Mongolia has been in the lowest-value area for a long time. Hebei is in the lower-value area most of the time. Tianjin and Shanxi changed greatly and finally jumped to the highest-value area and the higher-value area, respectively.

We use the standard deviation ellipse to identify the direction of TM in each province. As shown in Fig.3, the lengths of the minor semiaxis and major semiaxis of the ellipse increased significantly. The growth of the short semiaxis reveals that the degree of dispersion of TM in China's provinces is gradually increasing. This result is consistent with the previous analysis conclusions that TM in some provinces shows a more obvious transition trend, which makes the overall dispersion of TM increase.

Spatial correlation characteristics of tourism mobility

Global spatial autocorrelation of tourism mobility. We use ArcGIS 10.8 to calculate the global Moran's I of TM for 2000–2018, and the results are shown in the table below (Table 2). The global Moran's I values from 2000 to 2018 were all positive, and the results from 2000 to 2015 were not significant, and the results from 2016 to 2018 were all significant at the 90% level. TM presents a significant positive spatial correlation. This shows that provinces with high TM in China have relatively high TM in their surrounding areas. From the overall trend,

Year	Moran's I	Z- value	P-value	Year	Moran's I	Z -value	P-value
2000	0.015	0.876	0.381	2010	0.005	0.662	0.508
2001	0.009	0.710	0.478	2011	0.021	0.889	0.374
2002	0.024	0.948	0.343	2012	0.021	0.878	0.380
2003	0.035	1.139	0.255	2013	0.027	0.968	0.333
2004	0.062	1.430	0.153	2014	0.058	1.413	0.158
2005	0.075	1.615	0.106	2015	0.071	1.601	0.109
2006	0.026	0.874	0.382	2016	0.083	1.783	0.075*
2007	0.052	1.239	0.215	2017	0.081	1.745	0.081*
2008	0.063	1.409	0.159	2018	0.076	1.671	0.095*
2009	0.020	0.796	0.426				

Table 2. Global Moran's I index of TM.





https://doi.org/10.1371/journal.pone.0275605.g004

the spatial correlation degree of China's TM has gradually increased, but its value has not exceeded 0.1, indicating that the spatial agglomeration effect of China's TM is still weak.

Local spatial autocorrelation cluster of tourism mobility. The global Moran's I cannot reflect the spatial correlation exhibited by local regions or individual provinces. We further use ArcGIS 10.8 to draw the LISA cluster diagram for 2000, 2005, 2010, 2015, and 2018 (Fig 4). The research samples are divided into four types of agglomeration: provinces with high TM are surrounded by provinces with high TM (H-H agglomeration), provinces with high TM are surrounded by provinces with low TM (H-L agglomeration), provinces with low TM are surrounded by provinces with high TM (L-H agglomeration), and provinces with low TM are surrounded by provinces with low TM (L-L agglomeration).

The results show that (1) provinces with H-H aggregation of TM in different periods are relatively stable; L-L and L-H aggregation types are stable but mixed with changes; The H-L aggregation type does not appear, which indicates that there is no "darkness under the light" area for China's provincial TM. Provinces with high TM can improve the TM of weekly provinces to a certain extent. (2) The H-H agglomeration is mainly concentrated in Jiangsu and Zhejiang. These regions are economically developed and have high per capita discretionary income. Moreover, the tourism infrastructure in these regions is more complete than that in other regions, and the tourist reception scale is also higher, so their TM shows a high local concentration. (3) The L-L agglomeration types are mainly distributed in geographically remote areas such as Qinghai, Tibet, Gansu, and Xinjiang in inland China. Moreover, Xinjiang and Gansu temporarily withdraw from the L-L agglomeration area. The main reason for this pattern is that the transportation infrastructure in the areas above mentioned is relatively underdeveloped. The "space-time compression effect" brought about by the rapid development of China's transportation is not significant. Furthermore, due to the distance from the main tourist source markets, although the TM shows a high growth rate, it is still in the lowest-value area and the lowervalue area for a long time. (4) L-H agglomeration is mainly transferred in Anhui, Shandong and Hebei, and these provinces are located in the "Leading Area". The average value of TM in the surrounding provinces is generally high, forming a "collapse area" for TM.

The impact of tourism mobility on tourism economic growth Spatial autocorrelation of tourism economic growth

In this study, a Monte Carlo simulation was selected to analyze the spatial autocorrelation of TEG (Table 3). Moran's I was positive from 2000 to 2018. They passed the significance test of different

Year	Moran's I	Z-Score	P-Score	Year	Moran's I	Z-Score	P-Score
2000	0.117	1.405	0.086*	2010	0.317	3.177	0.005***
2001	0.113	1.406	0.087*	2011	0.307	2.989	0.007***
2002	0.110	1.233	0.083*	2012	0.241	2.401	0.009***
2003	0.109	1.428	0.080*	2013	0.215	2.312	0.017**
2004	0.117	1.335	0.086*	2014	0.234	2.273	0.006***
2005	0.118	1.455	0.082*	2015	0.249	2.535	0.008***
2006	0.105	1.188	0.101	2016	0.262	2.666	0.007***
2007	0.101	1.210	0.100*	2017	0.324	3.147	0.002***
2008	0.111	1.302	0.096*	2018	0.349	3.506	0.001***
2009	0.281	2.793	0.008***				

Table 3. Moran's I index of TEG.

Note

***, **, * indicate passing the significance test at the 1%, 5%, and 10% levels, respectively.

https://doi.org/10.1371/journal.pone.0275605.t003

degrees except in 2006, indicating that TEG has a significant positive spatial correlation. Therefore, a spatial econometric model should be selected to analyze the influencing factors of TEG.

Traditional panel data model

The unit root test using LLC and Fisher showed no unit root for TEG, TM, TH, TP, and TA (Table 4). The Kao test, Pedroni test, and Westerlund test were used to determine the cointegration relationship between the variables. The test results showed a cointegration relationship, indicating that the data can be used for modeling.

In terms of the regression model, the BP Lagrangian test results show the rejection of the mixed model. Wooldridge and Wald's test indicates the presence of heteroskedasticity and autocorrelation in the data. The presence of heteroskedasticity would lead to an increase in the variance of the model parameters and invalidate the Hausman test results. If the regression is still performed using the method without heteroskedasticity, it will undermine the validity of the t-test and F-test, while autocorrelation will exaggerate the significance of the parameters. Therefore, the panel model is selected by the over-identification test (Hausman test result is significant), and the result shows that the Sargan-Hansen statistic is 14.32 and significant, so fixed effect modeling should be selected.

Table 4. Results of stationarity test.

Unit root test	TEG		ТМ		ТР	TH		TA		
LLC test	-4.6610***		-19.0978**	*	-4.1181***	-6.8463***		-9.0320***		
Fisher test	139.3618**	39.3618***			88.8039**	98.7588***	98.7588***		92.5520***	
					Cointegration tests					
Kao test		Pedron test		Westerlu	Westerlund test					
Modified Dickey-Ful	Modified Dickey-Fuller t -2.0376**		Modified Phillips-Perron t		5.6168**	Variance	Variance ratio 2.6208***			
Dickey-Fuller t -3.5593***		Phillips-Perron t		-7.6942***						
Augmented Dickey-Fuller t -2.2142**										
Unadjusted modified Dickey-Fuller t -2.6291***			Augmented Dickey-Fuller t		-4.7115***					
Unadjusted Dickey-Fuller t -3		-3.8579***]						

Note

***, **, * indicate passing the significance test at the 1%, 5%, and 10% levels, respectively.

Variables	Fixed effects mode	Fixed effects model				Random effects model		
	Coefficients	Driscoll-Kı	aay standaro	l errors	Coefficient	s	Driscoll-Kraay standard errors	
LnTM	0.6186***	0.0341			0.4525		0.3299	
LnTP	0.2341***	0.0308	0.0308				0.0370	
LnTH	0.3276***	0.0526	0.0526				0.0625	
LnTA	0.5250***	0.0197	0.0197				0.0230	
_cons	-9.3119***	0.1385			-3.0771**		1.2723	
R-squared	0.9441				0.9234	234		
F-score	6982.81***			Wald test			5040.52***	
F-test	25.25***	25.25***			Hausman Test		26.68***	
Wooldridge test	5.692**	5.692**			Wald test		19391.36***	
Sargan–Hansen test	14.32***	14.32***			BP Lagrange multiplier test		433.09***	

Table 5. Regression test results.

Note

***, **, * indicate significance at the 1%, 5%, and 10% levels, respectively.

https://doi.org/10.1371/journal.pone.0275605.t005

To further address heteroskedasticity and autocorrelation, this study uses Driscoll-Kraay standard errors for regression. The results in <u>Table 5</u> show a significant positive effect of each variable on TEG, where each 1% increase in TM will promote 0.62% growth in the tourism economy.

Spatial panel data model

In this paper, the specific form of the spatial panel data model was determined by LM-LAG and LM-ERROR tests. If the result of LM-lag is significant and LM-error is not significant, then SLM should be used, and vice versa, SEM should be used. If LM-lag and LM-error statistics are significant, it indicates that the spatial correlation of the lag term and the spatial correlation of the residuals should be considered. In this case, the SDM can be used to set the model. Subsequently, this study determined whether the SDM model would degenerate into SLM or SEM by Wald and LR tests, and the results showed that all passed the significance test. Meanwhile, the test results of LM-lag, LM-error, LM-lag (robust), and LM-error (robust) were significant (Table 6), indicating that the model set using SDM has a certain rationality.

We selected the regression model through the Hausman test, and the result showed that the value was 19.31, and the corresponding probability value was 0.007, which indicated that the null hypothesis of random effect was rejected. Therefore, the fixed-effect model was selected for regression analysis. Table 7 shows the estimation results, where ρ rejects the original hypothesis only in the Spatio-temporal fixed-effects model. Therefore, this paper provides a

Statistical quantities	Value	Statistical quantities	Value				
LM-lag	1355.52***	LR-lag	42.17***				
LM-error	3519.48***	LR-error	126.61***				
LM-lag (robust)	6294.22***	Wald-lag	42.58***				
LM-error (robust)	8458.17***	Wald-error	48.83**				

Table 6. Tests of the spatial panel data model.

Note

***, ** indicate passing the significance test at the 1% and 5% levels, respectively.

Variables	Spatial fixed effects	Temporal fixed effects	Spatio-temporal fixed effects
LnMobility	0.512***	0.352***	0.482***
LnTP	-0.020	0.635***	-0.032
LnTH	0.114	0.348***	0.140**
LnTA	0.263***	0.055*	0.264**
Wx LnTM	-0.400**	0.843***	-0.449
Wx LnTP	0.506*	0.286	0.993***
Wx LnTH	1.081***	0.577**	1.693***
Wx LnTA	-0.129	-0.655***	-0.215
R ²	0.9604	0.900	0.9602
Spatial rho	-0.102	-0.641	-0.559***
Variance sigma2_e	0.030***	0.100***	0.031***

Table 7. Estimation results of SDM.

Note

***, **, * indicate significance at the 1%, 5%, and 10% levels, respectively.

https://doi.org/10.1371/journal.pone.0275605.t007

specific analysis of the Spatio-temporal fixed-effects model. The regression results indicate that TM shows a significant positive effect on regional TEG.

According to the results and spatial effect decomposition (Table 8), ρ is -0.559, indicating that the growth of the tourism economy in neighboring provinces will have a negative impact on the local area. The direct effect of TM is significant, indicating that TM will promote TEG. However, the indirect effect results show that the increase in TM in neighboring provinces will have a negative impact on the local TEG.

Decomposition of the influencing factors by LMDI. We decompose the influencing factors and analyze their contribution trend. <u>Table 9</u> shows the specific contribution of each influencing factor to the TEG in the seven regions.

 Δ T increases from 15.41% in 2000~2005 to 22.55% (2005~2010), and then decreases to 9.35% in 2010~2015 and 7.01% in 2015~2018. Overall, the Δ T showed a downward trend, but it is still an important factor in promoting TEG. The average contribution rate of the Δ T from 2000 to 2018 reached 14.82%.

 Δ I maintained an overall downward trend during 2000 ~2018. It gradually decreased from 31.42% (2000~2005) to 22.94% (2015~2018). In contrast, the added-value of tertiary industry per capita increases from 3653 yuan to 34,969 yuan in the same period, indicating that the contribution of tertiary industry to TEG continues to decline, and tourism is gradually decoupled from the development of the tertiary industry.

 Δ S maintained an overall upward trend during 2000~2018, from 7.09% (2000~2005) to 14.67% (2015~2018). The overall contribution rate was 11.50%, indicating that increasing the proportion of the tertiary industry in tourism can promote TEG.

Table 8. Decomposed spatial effects of SDM.

Variables	Direct effect	Indirect effect	Total effect
LnTM	0.497***	-0.486*	0.011
LnTP	-0.051	0.685***	0.634***
LnTH	0.104*	1.100***	1.204***
LnTA	0.275***	-0.241	0.035

Note

***, * indicate passing the significance test at the 1% and 10% levels, respectively.

Time period	Factors	North China	Northeast China	East China	Central China	South China	Southwest China	Northwest China	Nationwide
2000~2005	ΔTM	15.76%	14.84%	6.35%	3.95%	21.23%	17.87%	11.19%	12.54%
	ΔT	10.16%	11.75%	21.44%	20.95%	7.01%	11.58%	19.95%	15.41%
	ΔI	35.89%	26.88%	29.48%	36.28%	37.86%	28.57%	28.48%	31.42%
	ΔS	1.76%	20.29%	9.02%	3.99%	-6.94%	10.16%	8.99%	7.09%
	ΔR	-23.09%	-26.24%	-25.13%	-25.52%	-26.12%	-28.71%	-28.88%	-26.22%
2005~2010	ΔTM	-0.49%	17.41%	7.13%	3.21%	13.33%	7.27%	8.46%	7.35%
	ΔT	26.00%	14.29%	23.40%	24.65%	18.20%	24.22%	22.53%	22.55%
	ΔI	38.57%	23.43%	28.61%	21.69%	30.61%	24.63%	28.97%	28.66%
	ΔS	0.08%	13.68%	10.29%	20.83%	0.64%	10.78%	10.37%	9.15%
	ΔR	-17.25%	-31.18%	-28.49%	-27.56%	-29.49%	-31.09%	-29.26%	-27.49%
2010~2015	ΔTM	26.47%	-1.81%	22.39%	24.67%	21.74%	18.64%	16.93%	19.38%
	ΔT	5.98%	10.47%	7.47%	7.27%	9.43%	12.45%	12.77%	9.35%
	ΔI	23.73%	36.18%	28.55%	29.36%	25.01%	25.94%	26.25%	27.46%
	ΔS	14.19%	-3.96%	5.48%	6.77%	14.38%	12.42%	16.08%	9.79%
	ΔR	-28.85%	-8.71%	-28.48%	-31.15%	-29.45%	-29.85%	-27.98%	-27.12%
2015~2018	ΔTM	25.19%	27.07%	25.66%	21.80%	0.42%	16.34%	28.53%	21.87%
	ΔT	4.86%	5.34%	4.73%	8.61%	9.44%	13.65%	4.32%	7.01%
	ΔI	24.02%	13.12%	30.57%	28.73%	19.70%	23.41%	15.06%	22.94%
	ΔS	13.49%	21.26%	9.39%	10.02%	10.73%	18.48%	20.61%	14.67%
	ΔR	-29.64%	-33.20%	-28.61%	-29.39%	-7.86%	-28.11%	-31.48%	-27.67%
Overall contribution through	ΔTM	17.53%	16.99%	14.66%	11.96%	17.92%	15.58%	15.98%	15.76%
2000~2018	ΔT	12.01%	12.19%	16.15%	17.44%	13.85%	15.75%	15.40%	14.82%
	ΔI	31.15%	26.95%	28.89%	28.00%	33.53%	25.03%	25.00%	28.18%
	ΔS	7.33%	14.84%	11.69%	13.60%	5.56%	13.22%	13.97%	11.50%
	ΔR	-25.30%	-29.03%	-28.61%	-29.01%	-29.14%	-30.43%	-29.64%	-28.67%

Table 9. Contribution rate of factors influencing regional TEG.

ΔTM declined from 12.54% (2000~2005) to 7.35% (2005~2010), increased to 19.38% (2010~2015), and then reached 21.87% (2015~2018). The contribution of TM to TEG is stable at 14.66%~17.92%, except for Central China (11.96%), with an overall contribution of 15.76%, indicating that TM has a catalytic effect on TEG, and enhancing TM is a crucial way to promote tourism development.

https://doi.org/10.1371/journal.pone.0275605.t009

 ΔR shows a negative effect on TEG, and the degree of adverse effect increases slowly from 26.22% to 27.67%. The overall contribution rate was 28.67%. Reception is defined as the ratio of the resident population to the number of tourists. This shows that on the premise that the permanent resident population remains basically unchanged, the contribution to TEG can be effectively increased by expanding the scale of tourists.

Discussion

Regression results of tourism mobility on tourism economic growth

This study briefly analyzes the regression results of the traditional and spatial panel data model. However, the spatial autocorrelation test results of TEG show an overall trend of fluctuating and increasing spatial correlation, especially with 2009 as the abrupt change point and a significant increase in the degree of agglomeration. Therefore, the article discusses the results of the spatial panel data model in detail, and the primary purpose of analyzing the traditional panel data model is to compare it with the spatial econometric results. The regression results of the spatial econometric model show that both TM and TA have a significant positive impact on TEG, which verifies the hypothesis we proposed above. This result is also consistent with Wu et al. [93] and Perboli et al. [94]. In contrast, TP and TH have no significant impact on TEG. However, previous studies have also shown that the spatial spill-over effect of tourism can significantly affect the TEG [27–29]. Therefore, the impact of TP and TH on TEG remains to be further confirmed.

According to the decomposition results, TM will promote the growth of the local tourism economy but will have a negative impact on neighboring provinces, which indicates a more obvious competition in tourism development among provinces. The increase in mobility in a particular place under a given number of tourists will lead to a diversion of tourists, which will have a negative impact on neighboring regions. Therefore, the tourism industry should also pay attention to the competitive situation in the surrounding areas. The development of tourism focus not only on improving local tourism mobility but also on neighboring areas. Both TP and TH manifest substantial spatial spillover effects. The increase in TP and TH in neighboring areas will produce positive effects, making local areas attach importance to the development of tourism resources and enhancing tourism attraction. TA has a significant positive contribution to TEG, which is consistent with the conclusion of Yang and Wong [27]. However, the spatial spillover effects of TA on TEG are not significant, which may be related to the fact that air traffic does not depend on adjacent spaces.

Analysis of influencing factors' contribution rate to tourism economic growth

TM and \DeltaTM. The Δ TM in North, Central, Southwest, and South China all show a trend of "falling and rising." It should be noted that the Δ TM in North China was negative from 2005 to 2010, mainly due to the significant decline in TM in Tianjin and Hebei. The improvement in the transportation infrastructure has a significant impact on TM in Central and Southwest China. The opening of high-speed railroads is a fundamental reason for the fluctuation in Δ TM. For South China, due to the implementation of the overnight visitor count statistics in the tourism statistics system of Guangdong in 2015~2018, the number of tourists decreased significantly compared to 2010~2015, which in turn led to a significant weakening of the Δ TM. In contrast to the regions mentioned above, the Δ TM in Northeast China shows a trend of "rising and falling" changes. From 2010 to 2015, the contribution of TM to TEG in Northeast China declined and was negative. The main reason is the overall decline of the regional economy in the Northeast region at this stage. In 2014 and 2015, the GDP growth rates of Northeast China were 4.23% and -0.84%, respectively, ranking second and last among the seven regions in China during the same period. At the same time, the Northeast region began to carry out statistical "squeeze water" at this stage, which caused obvious fluctuations in the scale of tourists. Therefore, the downturn in the regional economic environment and stricter tourism statistics have negatively affected the contribution of tourism mobility to tourism economic growth. However, since 2016, China has put forward the " all-for-one tourism" policy. Provinces began to pay more attention to the role of tourism in regional economic growth. All-for-one tourism policies and new management systems have led to the continuous improvement of TM in Northeast China from 2015 to 2018, and the contribution to TEG has increased significantly compared with 2010–2015. The Δ TM in East China gradually increased from 6.35% to 25.66%, which is related to the opening of the high-speed railroad network in 2010, leading to a significant increase in TM. Northwest China has made the tourism industry a key point for economic growth, and its tourist reception and transportation construction levels have been rapidly improved under the impetus of the all-for-one tourism strategy.

Traffic and ΔT . The contribution of ΔT to TEG generally shows a downward trend. However, during the same period, Traffic showed a gradual upward trend. In 2018, it increased by 258.72% compared with 2000. Among them, it increased by 35.61% from 2000 to 2005, increased by 91.36% from 2005 to 2010, increased by 24.83% from 2010 to 2015, and increased by 10.73% from 2015 to 2018. From this, it can be judged that there may be a "threshold" in the transportation infrastructure. When the stock of transportation infrastructure in China reaches a certain level, the accumulation of transportation infrastructure cannot improve the contribution to the TEG. The role of transportation infrastructure in influencing tourists' decisions and determining TM cannot be ignored. However, its contribution rate gradually decreases as transportation facilities are gradually improved and regional accessibility differences narrow. The ΔT is 14.82% during the examination period, in which the contribution rate of Traffic to TEG in East China (16.15%), Central China (17.44%), Southwest China (15.75%), and Northwest China (15.40%) is higher than that in North, Northeast and South China. This is mainly because Central China and East China are the regions with the largest passenger turnover in China. From 2000 to 2018, the average passenger turnover in Central China and East China was 118.988 billion person-kilometers and 84.595 billion person-kilometers, respectively. The Southwest China and Northwest China are among the regions with the fastest growth in passenger turnover in China, increasing by 3.13 times and 1.77 times respectively, ranking first and second in all regions.

Industry and ΔI . The tertiary industry consists of transportation, warehousing and postal industry, information transmission, real estate industry, financial industry, wholesale and retail industry, accommodation and catering industry, etc. Tourism is only a part of it. The per capita added value of the tertiary industry reflects the degree of development of the service industry in various regions, and this indicator has achieved a relatively large increase in terms of changing trends. It increased from 3,653 yuan in 2000 to 34,969 yuan, an increase of 8.57 times. The contribution of ΔI to TEG has gradually declined, mainly due to the slowdown in the growth rate of the per capita added value of the tertiary industry. The growth rate dropped from 91.30% in 2000–2005 to 34.35% in 2015–2018. The contribution of ΔI to TEG in North China, South China, Northwest China, and Southwest China is consistent with the national trend. Northeast China, East China, and Central China show different trends. Especially in the Northeast region, the contribution of ΔI to TEG has dropped significantly. The overall contribution rate of Industry reached 28.18%, indicating that the quality of tertiary industry development has a vital role in promoting TEG. ΔI is generally stable in East and Central China and declines significantly in Northeast China, which may be related to the deceleration of tertiary industry development, as the data show that the added-value of tertiary industry per capita in Liaoning, Heilongjiang, and Jilin increased by 93.04%, 75.15% and 90.43% from 2010 to 2015, while it only grew by 0.63%, 39.88% and 23.18% from 2015 to 2018. Central China was inconsistent with the overall national trend from 2005 to 2010. This is mainly due to the slow increase in the per capita added value of the tertiary industry during this period, ranking last in all regions. During this period, the industrial structure of Central China was still dominated by industry. In 2010, the average industrial added value accounted for 56.37% of GDP, the highest in all regions of the country. East China was inconsistent with the overall national trend in 2015–2018. The main reason is that the proportion of the tertiary industry in Fujian and Jiangxi in the region has not exceeded 50%, and there is a large room for optimization and improvement of the industrial structure. Therefore, the growth rate of the added value of the tertiary industry per capita exceeds the previous stage, and the contribution of ΔI to TEG is still rising.

Structure and ΔS . The share of tertiary industry in tourism in Beijing and Tianjin increased significantly from 2010 to 2018 compared to 2000, leading to the rapid growth of ΔS

in North China. The Δ S in Northeast China was -3.96% from 2005 to 2010, mainly since the growth rate of tertiary industry in Heilongjiang and Liaoning lagged behind that of the tourism industry. The Δ S in East, Central, and Southwest China is relatively stable, indicating that tourism and tertiary industry maintain a coordinated development. The Δ S in South China has achieved a shift from negative to positive growth. As the economic volume of Guangdong accounts for a large proportion in South China and the growth rate of tourism significantly lags behind the development rate of the tertiary industry, it leads to a low Δ S in South China from 2000 to 2010. The opening of high-speed rail provides new opportunities for tourism development, and the Δ S in South China gradually increased to 14.38% and 10.73% in 2010~2018. The Δ S in Northwest China has been increasing, which suggests that the tourism economy is the primary driver of tertiary industry growth. The continuous growth of the Δ S contribution to TEG is partially consistent with the findings of Chang et al. [95], De Vita and Kyaw [96], and Zuo and Huang [97]. The higher Structure is, the greater the contribution of Δ S to TEG. However, the literature above mentioned also pointed out that Δ S has a turning point. For example, Zuo and Huang [97] found that this value in China is 8.25%.

Reception and \Delta \mathbf{R}. The $\Delta \mathbf{R}$ has a negative impact on TEG. Zuo and Huang [97] used the ratio of tourist arrivals to the permanent resident population to characterize tourism specialization in a study evaluating China's tourism-oriented economic growth. Before reaching the inflection point of 30.34 (that is, the tourism reception effect value is 0.03), this indicator has a significant positive impact on TEG. From 2000 to 2018, the tourism reception effect value dropped from 1.47 to 0.11, still less than 0.03. Therefore, the results of our study also partially confirm the research of Zuo and Huang [97]. While expanding the scale of tourists, various regions should also pay attention to the "inflection point" of the Reception value. When the inflection point is reached, the larger the scale of tourists is, the smaller the contribution to the TEG. However, the ratio of regional population to tourist decreases from 1.47 to 0.11 during the period from 2000 to 2018, indicating that not only the number of tourists should be taken into account, but also the quality of the tourism and the per capita tourism consumption should be attached importance to the TEG. ΔR is relatively stable, among which the southwest and northwest China have the most significant negative contribution to the TEG, indicating that the growth rate of the number of tourists received in the above regions is higher than that of other regions.

Conclusions

This paper proposes the concept of TM based on the hydrodynamic equation, constructs an econometric model of TEG with TM as the core explanatory variable, explores the direct and indirect effects of TM on TEG, measures the specific contribution of each influencing factor using the LMDI decomposition, and draws the following conclusions.

1. The TM in China has maintained rapid growth for a long time. However, there are differences in the rate of growth in different regions. East China and North China are Leading Area, with the highest average tourism mobility, but the smallest average annual growth rate; Central China, South China, Northeast China, and Southwest China are Stable Area, with the middle average TM and average annual growth rate; Northwest China is Potential Area, which has the smallest average TM, but the largest average annual increase. The TM in each region only showed a significant positive spatial correlation in 2016–2018. The space-time pattern is constantly changing over time. The high-value areas and high-value areas of TM increased significantly, while the low-value areas and low-value areas decreased significantly. The local spatial autocorrelation results of TM are stable, and various agglomeration states are stably distributed in some provinces.

- 2. The regression results of the traditional panel data model and the spatial panel data model both show that TM has a significant positive effect on TEG. Under the premise of considering the spatial effect, the improvement of TEG in a province by TM will have a negative impact on the adjacent province.
- 3. Applying the LMDI decomposition method, the TEG is decomposed into *TM*, *Traffic*, *Industry*, *Structure*, and Reception. The results show that the contribution of TM and Structure to TEG showed an upward trend, with average annual contribution rates of 15.76% and 11.50%, respectively. It indicates that improving TM is a crucial way to promote tourism development. The contribution of the *Traffic* and *Industry* to TEG generally showed a downward trend, with average annual contribution rates of 14.82% and 28.18%, respectively. The Reception has a negative impact on the TEG, but it is still a positive contribution, with an average annual contribution rate of 28.67%. The five types of effects of TEG decomposition were different due to regional differences.

The main contributions of this study are as follows: (1) Based on fluid mechanics, we constructed an indicator of TM. We comprehensively consider the impact of tourist arrivals and transportation infrastructure on TEG, which is rarely proposed by scholars in the literature. Our research enriches the research on the influencing factors of TEG. (2) We analyze the influence of TM on TEG based on the econometric model, which highlights the importance of TM. Moreover, we found that TM has negative spatial overflow.(3) Based on the LMDI method, we decompose TEG into five major effects, rather than just considering traditional variables such as human input, capital input, and tourism resource input. Our study further enriches the research on the influencing factors of TEG.

Based on our findings above, we draw the following policy implications. To improve TEG, late-developing regions should improve TM by building large-scale tourism transportation infrastructure, promoting destination marketing to attract tourists, and paying attention to the possible negative effects of increased TM in neighboring regions. At the same time, the improvement of TM should be emphasized at different stages. The threshold effect of tourism transportation infrastructure should also be fully considered. After the transportation infra-structure reaches a certain stock, its contribution to TEG will decrease. At this time, expanding the scale of tourists should become the main tourism development policy.

There are still some limitations in this study. It is difficult to directly collect data on the inflow and outflow of tourist between certain provinces. Therefore, we only select inflow of tourists as the primary data and do not consider the influence of the tourists' outflow on TM. In fact, increased transport accessibility will not only expand the inflow of tourists but also affect the outflow of tourists. Therefore, the superposition effect of traffic and tourist inflow/ outflow should be considered comprehensively to improve the scientific rationality of TM measurement. This study lacks comparative studies across multiple countries. The research in our study may show differentiated findings for developed or less developed countries. When constructing the econometric model, we mainly consider TM as the core explanatory variable, and only select human input and capital input, and air traffic related to traffic as control variables from the perspective of the economic growth model. In the future, the theory and practice of TM will be further explored with multivariate data to form a more rigorous and systematic cognitive framework.

Supporting information

S1 Fig. Map of the seven regions. (DOCX)

S1 File. Research data. (XLSX)

Author Contributions

Conceptualization: Jun Liu, Yun Tong.

Data curation: Fan Yu.

Formal analysis: Fan Yu.

Funding acquisition: Jun Liu.

Investigation: Yun Tong.

Methodology: Jun Liu, Yun Tong.

Resources: Jun Liu.

Software: Mengting Yue.

Writing - original draft: Mengting Yue.

Writing - review & editing: Fan Yu.

References

- 1. WTCF. World Tourism Economy Trends Report; 2020 [cited 2020 Jan 10]. Available from: http://travel. cctv.com/2020/01/10/ARTIGAy04Rv3zmZ4e7WO0SEz200110.shtml.
- Li KX, Jin M, Shi W. Tourism as an important impetus to promoting economic growth: a critical review. Tour Manag Perspect. 2018; 26: 135–142.
- 3. UNWTO. Tourism and Covid-19;2022. Available from: https://www.unwto.org/tourism-covid-19;2022. Available from: https://www.unwto.org/tourism-covid-19;2022. Available from: https://www.unwto.org/tourism-covid-19-coronavirus.
- 4. Szivas E, Riley M, Airey D. Labor mobility into tourism. Ann Tour Res. 2003; 30: 64-76.
- 5. Liu A, Wall G. Planning tourism employment: a developing country perspective. Tour Manag. 2006; 27: 159–170.
- 6. Urry J. Global complexity. Cambridge: Polity Press; 2003.
- Cárdenas-García PJ, Sánchez-Rivero M, Pulido-Fernández JI. Does tourism growth influence economic development? J Travel Res. 2013; 54: 206–221.
- 8. Hannam K, Butler G, Paris CM. Developments and key issues in tourism mobilities. Ann Tour Res. 2014; 44: 171–185.
- 9. Kim YR, Williams AM, Park S, Chen JL. Spatial spillovers of agglomeration economies and productivity in the tourism industry: the case of the UK. Tour Manag. 2021; 82: 104201.
- 10. Ministry of Culture and Tourism. Basic information of the tourism market in 2020. Beijing: Ministry of Culture and Tourism, PRC; 2020.
- Prideaux B. Factors affecting bilateral tourism flows. Ann Tour Res. 2005; 32: 780–801. https://doi.org/ 10.1016/j.annals.2004.04.008 PMID: 32572283
- 12. Wang X, Huang S, Zou T, Yan H. Effects of the high speed rail network on China's regional tourism development. Tour Manag Perspect. 2012; 1: 34–38.
- Massidda C, Etzo I. The determinants of Italian domestic tourism: a panel data analysis. Tour Manag. 2012; 33: 603–610.
- 14. Li LSZ, Yang FX, Cui C. High-speed rail and tourism in China: an urban agglomeration perspective. International Journal of Tourism Research. 2019; 21: 45–60.
- 15. Pearce DG. Towards a geography of tourism. Ann Tour Res. 1979; 6: 245–272.
- 16. Gunn CA. Vacationscape: designing tourist regions. New York: Van Nostrand Reinhold; 1988.
- 17. Khadaroo J, Seetanah B. The role of transport infrastructure in international tourism development: a gravity model approach. Tour Manag. 2008; 29: 831–840.
- **18.** Chi J. A cointegration analysis of bilateral air travel flows: the case of international travel to and from the United States. J Air Transp Manag. 2014; 39: 41–47.

- Liu H, Song HY, Wang YL. Inbound tourism demand and economic growth in China–empirical study based on mixed frequency Granger causality tests. Econ Manag. 2016; 38: 149–160.
- Zhang Y, Li X, Wu T. The impacts of cultural values on bilateral international tourist flows: a panel data gravity model. Current Issues in Tourism. 2019; 22: 967–981.
- Saayman A, de Klerk J. Forecasting tourist arrivals using multivariate singular spectrum analysis. Tourism Economics. 2019; 25: 330–354.
- Keum K. Tourism flows and trade theory: a panel data analysis with the gravity model. Ann Reg Sci. 2010; 44: 541–557.
- 23. Morley C, Rosselló J, Santana-Gallego M. Gravity models for tourism demand: theory and use. Ann Tour Res. 2014; 48: 1–10.
- 24. Kadir N, Karim MZA. Tourism and economic growth in Malaysia: evidence from tourist arrivals from Asean-S countries. Econ Res. 2012; 25: 1089–1100.
- Leitã NC, Shahbaz M. Economic growth, tourism arrivals and climate change. In Bulletin of energy economics (BEE), the economics and social development organization (TESDO); 2016. pp. 35–43.
- 26. Zhang A, Liu L, Liu G. High-speed rail, tourist mobility, and firm value. Econ Model. 2020; 90: 108–116.
- Yang Y, Wong KKF. A spatial econometric approach to model spillover effects in tourism flows. J Travel Res. 2012; 51: 768–778.
- 28. Yang Y, Fik T. Spatial effects in regional tourism growth. Ann Tour Res. 2014; 46: 144–162.
- **29.** Yuming W. Spatial panel econometric analysis of tourism economic growth and its spillover effects. Tourism Tribune/Lvyou Xuekan. 2014; 29: 16–24.
- Carmona MJC, Collado RR. LMDI decomposition analysis of energy consumption in Andalusia (Spain) during 2003–2012: the energy efficiency policy implications. Energy Effic. 2015; 9: 807–823.
- Chong C, Liu P, Ma L, Li Z, Ni W, Li X, et al. LMDI decomposition of energy consumption in Guangdong Province, China, based on an energy allocation diagram. Energy. 2017; 133: 525–544.
- Pablo-Romero MDP, Sánchez-Braza A, Sánchez-Rivas J. Tourism and electricity consumption in 9 European countries: a decomposition analysis approach. Current Issues in Tourism. 2021; 24: 82–97.
- Zha J, Dai J, Ma S, Chen Y, Wang X. How to decouple tourism growth from carbon emissions? A case study of Chengdu, China. Tour Manag Perspect. 2021; 39: 100849.
- 34. Oppermann M. A model of travel itineraries. J Travel Res. 1995; 33: 57–61.
- 35. Song H, Witt SF. Forecasting international tourist flows to Macau. Tour Manag. 2006; 27: 214–224.
- **36.** Song H, Li G. Tourism demand modelling and forecasting—a review of recent research. Tour Manag. 2008; 29: 203–220.
- Saayman A, Botha I. Non-linear models for tourism demand forecasting. Tourism Economics. 2017; 23: 594–613.
- Durden GC, Silberman J. The determinant's of florida tourist flows: a gravity model approach. Rev Reg Stud. 1975; 5: 41.
- Lozano S, Gutiérrez E. A complex network analysis of global tourism flows. International Journal of Tourism Research. 2018; 20: 588–604.
- Hwang Y-H, Gretzel U, Fesenmaier DR. Multicity trip patterns: tourists to the United States. Ann Tour Res. 2006; 33: 1057–1078.
- 41. Wang D, Wang L, Chen T, Lu L, Niu Y, Alan AL. HSR mechanisms and effects on the spatial structure of regional tourism in China. Journal of Geographical Sciences. 2016; 26: 1725–1753.
- Liu Z, Lu C, Mao J, Sun D, Li H, Lu C. Spatial-temporal heterogeneity and the related influencing factors of tourism efficiency in China. Sustainability. 2021; 13: 5825.
- Lue C-C, Crompton JL, Fesenmaier DR. Conceptualization of multi-destination pleasure trips. Ann Tour Res. 1993; 20: 289–301.
- 44. Li X, Meng F, Uysal M. Spatial pattern of tourist flows among the Asia-Pacific countries: an examination over a decade. Asia Pacific Journal of Tourism Research. 2008; 13: 229–243.
- **45.** McKercher B, Lau G. Movement patterns of tourists within a destination. Tourism Geographies. 2008; 10: 355–374.
- Zheng W, Huang X, Li Y. Understanding the tourist mobility using GPS: where is the next place? Tour Manag. 2017; 59: 267–280.
- Connell J, Page SJ. Exploring the spatial patterns of car-based tourist travel in Loch Lomond and Trossachs National Park, Scotland. Tour Manag. 2008; 29: 561–580.
- Hallo JC, Beeco JA, Goetcheus C, McGee J, McGehee NG, Norman WC. GPS as a method for assessing spatial and temporal use distributions of nature-based tourists. J Travel Res. 2012; 51: 591–606.

- Smallwood CB, Beckley LE, Moore SA. An analysis of visitor movement patterns using travel networks in a large marine park, North-Western Australia. Tour Manag. 2011; 33: 517–528.
- Xia J, Zeephongsekul P, Packer D. Spatial and temporal modelling of tourist movements using Semi-Markov processes. Tour Manag. 2011; 32: 844–851.
- Pablo-Romero MDP, Molina JA. Tourism and economic growth: a review of empirical literature. Tour Manag Perspect. 2013; 8: 28–41.
- 52. Tugcu CT. Tourism and economic growth nexus revisited: a panel causality analysis for the case of the Mediterranean Region. Tour Manag. 2014; 42: 207–212.
- Lee C-C, Chien M-S. Structural breaks, tourism development, and economic growth: evidence from Taiwan. Math Comput Simul. 2008; 77: 358–368. https://doi.org/10.1016/j.matcom.2007.03.004 PMID: 32288113
- 54. Payne JE, Mervar A. Research note: the tourism–growth nexus in Croatia. Tourism Economics. 2010; 16: 1089–1094.
- 55. Odhiambo NM. Tourism development and economic growth in Tanzania: empirical evidence from the ARDL-bounds testing approach. Econ Comput Econ Cybern Stud Res. 2011; 45: 71–83.
- Melián-González A, García-Falcón JM. Competitive potential of tourism in destinations. Ann Tour Res. 2003; 30: 720–740.
- 57. Wang Y. Tourism resource endowment and regional tourism economies: a positive analysis on shanxi province. Ecol Econ. 2010; 228: 41–45.
- Zhu X, He Y. Does tourism promote economic growth in the ethnic areas of China? Emerging Markets Finance and Trade. 2019; 57: 386–399.
- Katircioglu ST, Feridun M, Kilinc C. Estimating tourism-induced energy consumption and CO2 emissions: the case of Cyprus. Renew Sustain Energy Rev. 2014; 29: 634–640.
- Nitivattananon V, Srinonil S. Enhancing coastal areas governance for sustainable tourism in the context of urbanization and climate change in Eastern Thailand. Advances in Climate Change Research. 2019; 10: 47–58.
- Hamaguchi Y. Does the trade of aviation emission permits lead to tourism-led growth and sustainable tourism? Transp Policy. 2021; 105: 181–192.
- Yong EL. Understanding the economic impacts of sea-level rise on tourism prosperity: conceptualization and panel data evidence. Advances in Climate Change Research. 2021; 12: 240–253.
- Fahimi A, Akadiri SS, Seraj M, Akadiri AC. Testing the role of tourism and human capital development in economic growth. A panel causality study of micro states. Tour Manag Perspect. 2018; 28: 62–70.
- 64. Elsharnouby TH, Elbanna S. Change or perish: examining the role of human capital and dynamic marketing capabilities in the hospitality sector. Tour Manag. 2021; 82: 104184.
- Lin L. The impact of service innovation on business performance: evidence from firm-level data in Chinese tourism sector. In ICSSSM11. Tianjin, China: IEEE; 2011. pp. 1–5.
- Feng Q, Sun G. Effects of per capita GDP and urbanization on domestic tourism development in China's eight regions. Areal Research and Development. 2016; 35: 92–98. Chinese.
- Fu X, Ridderstaat J, Jia H. Are all tourism markets equal? Linkages between market-based tourism demand, quality of life, and economic development in Hong Kong. Tour Manag. 2020; 77: 104015.
- Macchiavelli A, Pozzi A. Low-cost flights and changes in tourism flows: evidence from Bergamo-Orio Al Serio international. In: Pechlaner H, Smeral E, editors. Tourism and leisure: current issues and perspectives of development. Wiesbaden: Springer Fachmedien Wiesbaden; 2015. pp. 323–336.
- 69. Khan SAR, Qianli D, SongBo W, Zaman K, Zhang Y. Travel and tourism competitiveness index: the impact of air transportation, railways transportation, travel and transport services on international inbound and outbound tourism. J Air Transp Manag. 2017; 58: 125–134.
- 70. Zhong L, Sun S, Law R. Movement patterns of tourists. Tour Manag. 2019; 75: 318–322.
- Kanwal S, Rasheed MI, Pitafi AH, Pitafi A, Ren M. Road and transport infrastructure development and community support for tourism: the role of perceived benefits, and community satisfaction. Tour Manag. 2020; 77: 104014.
- Dritsakis N. Tourism as a long-run economic growth factor: an empirical investigation for Greece using causality analysis. Tourism Economics. 2004; 10: 305–316.
- Zuo B, Cai S, Yang Y, Bao J. The structural changes in the topological network and their effects on performance of the tourism industry: the case of Yangshuo, China. Tourism Tribune. 2020; 35: 25–39.
- Causevic S, Lynch P. Political (in)stability and its influence on tourism development. Tour Manag. 2013; 34: 145–157.

- Liu C, Dou X, Li J, Cai LA. Analyzing government role in rural tourism development: an empirical investigation from China. J Rural Stud. 2020; 79: 177–188.
- 76. Di Matteo D. Effectiveness of place-sensitive policies in tourism. Ann Tour Res. 2021; 90: 103146.
- 77. Sigala M. New technologies in tourism: from multi-disciplinary to anti-disciplinary advances and trajectories. Tour Manag Perspect. 2018; 25: 151–155.
- Tu J, Zhang D. Does tourism promote economic growth in Chinese ethnic minority areas? A nonlinear perspective. Journal of Destination Marketing & Management. 2020; 18: 100473.
- Sun J, Zhou S, Wang N, Zhu H, Zhou D, Zhen F, et al. Mobility in geographical research: time, space and society. Geographical Research. 2016; 35: 1801–1818.
- Neil L. Tourism systems: an interdisciplinary perspective. Palmerston North, NZ: Department of Management Systems, Business Studies Faculty, Massey University; 1990.
- 81. McKercher B. A chaos approach to tourism. Tour Manag. 1999; 20: 425–434.
- 82. Chew J. Transport and tourism in the year 2000. Tour Manag. 1987; 8: 83-85.
- Khadaroo J, Seetanah B. Transport infrastructure and tourism development. Ann Tour Res. 2007; 34: 1021–1032.
- Masson S, Petiot R. Can the high speed rail reinforce tourism attractiveness? The case of the high speed rail between Perpignan (France) and Barcelona (Spain). Technovation. 2009; 29: 611–617.
- 85. Prideaux B. The role of the transport system in destination development. Tour Manag. 2000; 21: 53-63.
- **86.** Liu Y, Shi J. How inter-city high-speed rail influences tourism arrivals: evidence from social media check-in data. Current Issues in Tourism. 2017; 22: 1025–1042.
- Nazneen S, Xu H, Din NU. Cross-border infrastructural development and residents' perceived tourism impacts: a case of China–Pakistan economic corridor. International Journal of Tourism Research. 2019; 21: 334–343.
- Chen Y, Xie B, Zhang A. The impact of traffic on spatial mobility at different scales. Acta Geogr Sin. 2018; 73: 1162–1172. Chinese.
- Lu W, Liu W, Hou M, Deng Y, Deng Y, Zhou B, et al. Spatial-temporal evolution characteristics and influencing factors of agricultural water use efficiency in Northwest China—based on a super-DEA model and a spatial panel econometric model. Water. 2021; 13: 632.
- Ghalhari GF, Roudbari AD. An investigation on thermal patterns in Iran based on spatial autocorrelation. Theor Appl Climatol. 2018; 131: 865–876.
- Ren H, Shang Y, Zhang S. Measuring the spatiotemporal variations of vegetation net primary productivity in Inner Mongolia using spatial autocorrelation. Ecol Indic. 2020; 112: 106108.
- 92. Chen J, Yang ST, Li HW, Zhang B, Lv JR. Research on geographical environment unit division based on the method of natural breaks (Jenks). The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences. 2013; 3: 47–50.
- Wu C, Hayashi Y, Funck C. The role of charter flights in Sino-Japanese tourism. J Air Transp Manag. 2012; 22: 21–27.
- Perboli G, Ghirardi M, Gobbato L, Perfetti F. Flights and their economic impact on the airport catchment area: an application to the Italian tourist market. J Optim Theory Appl. 2015; 164: 1109–1133.
- Chang CL, Khamkaew T, McAleer MJ. A panel threshold model of tourism specialization and economic development (No. El 2009–40). Erasmus School of Economics (ESE). 2009. Available from: <u>http://hdl.</u> handle.net/1765/17310.
- **96.** De Vita G, Kyaw KS. Tourism specialization, absorptive capacity, and economic growth. J Travel Res. 2016; 56: 423–435.
- Zuo B, Huang S. Revisiting the tourism-led economic growth hypothesis: the case of China. J Travel Res. 2017; 57: 151–163.